Author's response to reviews

Title: A typhoid fever outbreak in a slum of South Dumdum municipality, West Bengal, India, 2007: Evidence for foodborne and waterborne transmission

Authors:

Rama Bhunia (rbhunia@gmail.com)
Yvan Hutin (hutiny@searo.who.int)
Ramachandran Ramakrishnan (dramakrishnan.nie@gmail.com)
Nitish Pal (pal_nishith@yahoo.co.in)
Tapas Sen (tapassen56@rediffmail.com)
Manoj Murhekar (mmurhekar@gmail.com)

Version: 5 Date: 14 April 2009

Author's response to reviews:

We have made the formatting changes as suggested and the revised version is enclosed herewith.